Author's response to reviews

Title: In-hospital complications after invasive strategy for the management of Non STEMI: women fare as well as men

Authors:

Caroline Berthillot (caroline.berthillot@chru-strasbourg.fr)
Dominique Stephan (dominique.stephan@chru-strasbourg.fr)
Michel Chauvin (michel.chauvin@chru-strasbourg.fr)
Gerald Roul (geraldroul@aol.com)

Version: 10 Date: 22 June 2010

Author's response to reviews:

Dear Editor-in-Chief,

The manuscript has been reviewed in order to correct any misspelling one more time. We found some remaining typographic errors and these were corrected as suggested. The words corrected were: staphylococcus, creatinine and favorable. Nothing else was modified.

Best regards
G ROUL